Author's response to reviews

Title: Validation of diabetes mellitus and hypertension diagnosis in computerized medical records in primary health care

Authors:

Carmen de Burgos-Lunar (carmenblunar@hotmail.com)
Miguel A Salinero-Fort (Miguel.salinero@salud.madrid.org)
Juan Cárdenas-Valladolid (jcardenas.gapm04@salud.madrid.org)
Sonia Soto-Díaz (sosoto.gapm04@salud.madrid.org)
Carmen Y Fuentes-Rodríguez (carmeny.fuentes@salud.madrid.org)
Juan C Abánades-Herranz (jabanades.gapm04@salud.madrid.org)
Isabel del Cura-González (icura.gapm09@salud.madrid.org)

Version: 6 Date: 26 October 2011

Author's response to reviews: see over
Dear Editor and production team,

We have made two corrections of typographical errors:

- We have replaced "CRD" by "CCR" in the first paragraph of the Background section
- We have replaced "HT" by "HTN" in the first paragraph of the Methods section

Kind regards,

In the name of the authors,

Carmen de Burgos Lunar,

Hospital Carlos III
Calle Sinesio Delgado, 10
28029 Madrid, Spain